Author’s response to reviews

Title: Factors associated with the early termination of exclusive breastfeeding among mother-infant dyads in Samara-Logia, Northeastern Ethiopia

Authors:

Amanuel Beyene (amanuel0216@gmail.com)

Misgan Liben (lmisgan@yahoo.com;lmisganl@gmail.com)

Amit Arora (A.Arorea@westernsydney.edu.au)

Version: 4 Date: 20 Oct 2019

Author’s response to reviews:

Thank you for pointing out this.

Dear editor, written informed consent was taken from the study participants. This is an editorial problem.